



BIS Working Papers No 1189

Sixty years of global inflation: a post GFC update

by Raphael Auer, Mathieu Pedemonte and Raphael Schoenle

Monetary and Economic Department

May 2024

JEL classification: E31, E52, E58, F02, F41, F42, F14, F62.

Keywords: globalization, inflation, Phillips curve, monetary policy, global value chain, international inflation synchronization. BIS Working Papers are written by members of the Monetary and Economic Department of the Bank for International Settlements, and from time to time by other economists, and are published by the Bank. The papers are on subjects of topical interest and are technical in character. The views expressed in them are those of their authors and not necessarily the views of the BIS.

This publication is available on the BIS website (www.bis.org).

© Bank for International Settlements 2024. All rights reserved. Brief excerpts may be reproduced or translated provided the source is stated.

ISSN 1020-0959 (print) ISSN 1682-7678 (online)

Sixty Years of Global Inflation: A Post-GFC Update^{*}

Raphael Auer[†]

Mathieu Pedemonte[‡]

Raphael Schoenle[§]

May 2024

Abstract

Is inflation (still) a global phenomenon? We study the international co-movement of inflation based on a dynamic factor model and in a sample spanning up to 56 countries during the 1960-2023 period. Over the entire period, a first global factor explains approximately 58% of the variation in headline inflation across all countries and over 72% in OECD economies. The explanatory power of global inflation is equally high in a shorter sample spanning the time since 2000. Core inflation is also remarkably global, with 53% of its variation attributable to a first global factor. The explanatory power of a second global factor is lower, except for select emerging economies. Variables such as a broad dollar index, the US federal funds rate, and a measure of commodity prices positively correlate with the first global factor. This global factor is also correlated with US inflation during the 70s, 80s, the GFC, and COVID. However, it lags these variables during the post-COVID period. Country-level integration in global value chains accounts for a significant proportion of the share of both local headline and core inflation dynamics explained by global factors.

JEL classification: E31, E52, E58, F02, F41, F42, F14, F62

Keywords: globalization, inflation, Phillips curve, monetary policy, global value chain, international inflation synchronization

^{*}This paper was prepared for the Research Handbook of Inflation. We thank the editors Guido Ascari and Riccardo Trezzi, as well as Ryan Banerjee, Enisse Kharroubi, Benoit Mojon and seminar participants at the BIS for comments and suggestions. We thank Kelsey O'Rourke for excellent research assistance. The views presented are not necessarily those of the BIS, the Federal Reserve Bank of Cleveland, or the Federal Reserve System.

[†]Bank for International Settlements and CEPR, raphael-auer@bis.org

[‡]Federal Reserve Bank of Cleveland, mathieu.pedemontelavis@clev.frb.org

[§]Brandeis University and CEPR, schoenle@brandeis.edu

1 Introduction

The influence of global factors on local inflation processes has significant implications for policy design and economic stability. The importance of global inflationary forces may challenge traditional monetary policy frameworks that rely mostly on tools that focus on domestic economic conditions, making them less effective. Therefore, the increasing importance of these global forces calls for a broader perspective that considers international conditions in the design of domestic policies.¹ These considerations are particularly relevant for smaller and more open economies.

To what extent has the recent inflation experience been a global phenomenon? Building on the empirical approach of Ciccarelli and Mojon (2010),² this article dissects the international co-movement of inflation using a dynamic factor model, with a focus on different periods of globalization. We analyze three samples: one starting in 1960, one in 1980, and one in 2000, all extending until June 2023. A particular contribution of our analysis to the work of Ciccarelli and Mojon (2010) thus lies in examining the importance of common factors in global inflation dynamics in the aftermath of two large global shocks: the post-GFC years and the recent post-pandemic inflation episode.³

As our analysis shows, the first and second factors associated with inflation dynamics exhibit several pronounced characteristics over different periods of globalization: independently of the period considered, the first factor appears relatively similar across samples and explains a large share of the variation of inflation across countries, around 58% in the full sample from 1960-2023 and over 72% in OECD economies. The second factor is relatively unstable across periods. It mostly captures country-specific dynamics, explaining little of the total variation, around 3%-5%. The importance of factors also varies across the development dimension: The importance of the first factor for explaining inflation is high and stable for advanced economies (AEs). For emerging market and developing economies (EMDEs), its im-

¹See, for example, Borio and Filardo (2007); Bianchi and Civelli (2015); Auer and Mehrotra (2014); Wynne (2012); Auer, Borio, and Filardo (2017); Forbes (2019); and Ascari and Fosso (2024).

²See also related approaches in Mumtaz and Surico (2009, 2012).

³See also Ha et al. (2023) for recent evidence on international inflation synchronization. Gilchrist et al. (2017, 2023) explicitly highlight financial frictions as a key feature of inflation dynamics during the GFC and the euro crisis.

portance changes over time. For example, its importance is low for countries such as Peru or Argentina in the 1960s but increases in later periods. Because the first global factor explains a substantial share of inflation in many economies, the second factor captures other local inflation experiences, mostly in EMDEs. The two factors also matter in a heterogeneous fashion for the world relative to the experience of the US: During the post-pandemic inflation surge, US inflation tends to lead both factors while in earlier periods global inflation dynamics are more synchronized with the US inflation experience. We also examine the relationship between inflation in China and global factors, finding in particular that it is mildly associated with the second global factor.

When we regress several economic variables on the first and second global factors generated by the factor model, several time-series correlates of global inflation emerge: Commodity prices and the VIX are positively correlated with both factors, but the VIX is only statistically significant for the second factor. In contrast, the US dollar index and the US federal funds rate are positively correlated with the first factor, while they are negatively correlated with the second factor. The factors thus capture various shocks that drive costs and lead to inflationary pressure, but they also capture the role of US monetary policy and general market volatility.

To assess the impact of global value chains (GVCs) on the importance of the *factors* underlying inflation dynamics, we estimate the relationship between the share of inflation explained by our two factors and the share of foreign value added in each country, which is a measure of the input-intensity of local production. GVC measures are significantly positively related to the share of local inflation dynamics that are explained by the first factor of our analysis: Countries with a higher share of their local inflation explained by the first global factor also have a higher share of foreign value added. Moreover, we find that the second factor does not bear any economically or statistically significant relationship with GVCs. It simply has low explanatory power, which suggests that little of this factor is explained by global trade factors.

Our results are consistent with the hypothesis that global shocks are an important driver of national inflation dynamics, but they do not necessarily imply such causality. For example, central banks could have simply been relatively more successful in controlling local shocks, keeping inflation overall low and stable over the past 30 years. While policies to control local shocks are relatively well studied, dealing with international supply shocks is challenging, and under certain conditions, an *accommodative* monetary policy can be desirable (Blinder, 1981), especially when we consider second round effects and other frictions (Bandera et al., 2023). Our results suggest that this challenge remains today, especially for economies more exposed to international price shocks, as most deviations from their target are explained by a common international factor.

Literature A large literature has studied the drivers and dynamics of inflation. Two strands are closely related to the analysis of this chapter. First, a series of papers argues in favor of an international component of inflation. For example, Ha, Kose, and Ohnsorge (2019) find that inflation in advanced economies is more strongly influenced by global factors than inflation in EMDEs. Additionally, they find that the global factor has become more important over time, specially for EMDEs. Likewise, Ascari and Fosso (2024) provide evidence for the increasing importance of global factors based on a Bayesian VAR analysis. Our analysis complements these papers by looking at common factors in AEs and EMDEs using the methodology of Ciccarelli and Mojon (2010), for different periods of globalization, and the relevance of economic variables that drive the factors (see Ha et al. (2023) for a contemporaneous study examining the drivers of global inflation).

Similarly, a series of papers has examined the role of economic factors as drivers of global inflation. The rise in commodity prices has often been cited as an inflationary force that transcends borders, given the global nature of commodity markets, as in De Gregorio (2012) and Peersman (2022). The trade integration of some economies, such as China and other emerging markets, to the global economy can also play a key role, by having a relevant disinflationary effect on prices of manufactured goods worldwide (see, for example, Auer and Fischer (2010)). A key aspect to consider in this trade context is, in particular, the role of GVCs: As production processes have become more global, the propagation of cost changes in one part of the world can quickly affect prices elsewhere. The influence of global value chains on inflation dynamics underscores the complexity of tracing inflationary impulses in a globalized economy. This channel has received substantial attention in the discussion of how prices spill over internationally, for example, in Auer, Borio, and Filardo (2017); Auer, Levchenko, and Sauré (2019). Our analysis explicitly relates our estimated factors to commodity price movement and its role in explaining the relationship between local inflation and the local importance of GVCs.

A second strand of the literature related to our analysis highlights the

role of monetary policy as a driver of co-movement in international inflation. Not only does the increasing importance of global forces challenge traditional monetary policy frameworks that rely mostly on domestic economic conditions, but the global reach of the monetary policies of key central banks, such as the US Federal Reserve and the European Central Bank, has also an influence on the importance of global factors (Miranda-Agrippino and Rey (2020); Degasperi, Hong, and Ricco (2020)). This influence can be particularly important for emerging economies (Kalemli-Özcan and Unsal, 2024; Kalemli-Özcan, 2019), depends on the local policies (Obstfeld, Ostry, and Qureshi, 2019) and can influence emerging markets indirectly, for example through inflation expectations (Frache et al., 2023). The global influence of the policy actions of major central banks can be a result of financial linkages (Rey (2015); Bruno and Shin (2015)) and currency pricing choices of exports and imports, as discussed in the dominant currency pricing literature (Gopinath et al., 2020; Mukhin, 2022; Egorov and Mukhin, 2023).

At the same time, a few papers also argue that monetary policy has played an important role in stabilizing inflation around sustainable boundaries for quite some time. For example, McLeay and Tenreyro (2020) argue that the apparent weakening of a short-run relationship between inflation and domestic measures of slack may simply be due to the optimal conduct of monetary policy. In a similar way, Bergholt, Furlanetto, and Vaccaro-Grange (2023) argue that monetary policy has shifted to a more firm commitment to maintaining price stability. These findings align with our interpretation of the second factor as a non-global component, which does not account for much of inflation variability, and an important first factor.

The structure of the paper is as follows: Section 2 discusses our data and methodology. Section 3 describes and analyzes the patterns of global inflation. Section 4 analyzes the economic forces behind global inflation factors. Section 5 concludes.

2 Data and Methodology

The subsequent analysis is based on monthly consumer price index (CPI) data from the BIS Data Portal for a panel of as many as 56 countries. The choice of time frames for the analysis differs, starting as early as 1960 and extending until June 2023. Depending on this choice, the set of available countries changes. A trade-off embodied in the choice of countries thus lies

in the length of the time series and the size of the panel. The subsequent specification with the most data will use data starting in 1999 and comprising 56 countries.

Our analysis focuses on headline inflation, since it is the main policy target in most economies and also widely available. More specifically, the focus on headline CPI inflation derives from most central banks choosing to set their policy objectives by targeting this measure of CPI inflation, in line with theoretical models. In some countries, however, central banks target core inflation which mainly removes some tradable components such as commodities prices and food prices. We thus also consider robustness exercises that use core inflation in a subsample of countries. We note that analysis of headline inflation (rather than just core inflation or some other subsets) is useful even for countries targeting core inflation: The analysis of headline inflation may indeed explain why some central banks avoid considering tradable components, as they are hard to control via traditional monetary policy tools.

Before discussing our main approach based on a factor model, we describe the evolution of raw inflation dynamics. To connect our discussion to some of the later insights as well as key aspects of globalization, we do so separately for AEs and EMDEs.⁴

This initial comparison is, moreover, based on two measures of inflation. The contrast of the two measures is useful because it further brings out a key characteristic of global inflation dynamics.⁵ As a first measure, we compute *simple averages* of inflation for both AEs and EMDEs. As a second measure, we also construct a *normalized measure* of inflation that we then average across countries. To construct this measure, we subtract from the inflation rate of each country its mean inflation rate and divide by the relevant country's standard deviation of inflation over the sample. After this procedure, we take simple averages across countries in each set. This normalization effectively reduces the weight of high-volatility, high-inflation-level economies.

⁴Table 7 in the Appendix contains the list of AE and EMDE countries. We classify countries following definitions from the Bank of International Settlements, see https://www.bis.org/statistics/country_groupings.pdf.

⁵Due to economic growth over time and reclassification of countries, we focus on a rather short time series for this particular comparison starting in 2000.

Figure 1: Average Inflation and Average Normalized Inflation for Advanced Economies and Emerging Market and Developing Economies (EMDEs)



Notes: The panels in the figure show monthly inflation trajectories for advanced economies (AEs) and emerging market and developing economies (EMDEs). Table 7 in the Appendix contains the list of advanced economies and EMDEs. The left panel shows simple averages and the right panel shows averages of a standardized inflation where each country's local inflation is demeaned and normalized by the standard deviation.

Figure 1 illustrates two salient patterns of global inflation. The figure shows the resulting time series for the *simple average* in its left panel and for the *normalized average* in its right panel. First, a characteristic that emerges clearly from the left panel of Figure 1 is that, on average, EMDEs have experienced a higher rate of inflation compared to AEs. In particular, at the beginning of the 2000s, EMDEs were experiencing inflation rates above 10% on average, while AEs were experiencing low rates of inflation. High inflation in EMDEs overall is influenced by high inflation in some Latin American countries, such as Argentina, and some eastern European economies, while other countries such as China experienced low inflation during that period and provided a downward weight on the high average rate in EMDEs.

Second, a further feature of the inflation experience across countries lies in a strong global co-movement evident since the early 2000s. This feature comes out very clearly once the normalized inflation measure is used to characterize the inflation dynamics, shown in the right panel of Figure 1. This normalization effectively reduces the weight of high-volatility, high-inflation economies. As a result, the dynamics between developed and developing economies are remarkably similar up to 2012. Thereafter, AEs experienced a more intense disinflation, due in part to the crisis in the Eurozone, followed by a relatively delayed inflation rise after COVID in the set of EMDEs. Conditioning on the volatility and level of the inflation experiences of countries highlights that overall inflation dynamics look similar across these two groups of AEs and EMDEs.

To investigate the degree of inflation co-movement more formally, our main approach gauges the strength of such co-movement in the global economy using a common-factor analysis as in Ciccarelli and Mojon (2010). Ciccarelli and Mojon (2010) use a factor representation based on Forni et al. (2000) and Stock and Watson (2002) that decomposes inflation rates as follows:

$$\Pi_t = \Lambda f_t + \varepsilon_t \tag{1}$$

where Π_t denotes an $n \times 1$ vector of inflation for the panel of countries. A captures the response of each country's inflation rate to the common factor f_t . Our analysis assumes normality of the error term ε_t and orthogonality with f_t .

Our analysis focuses on the first two factors, for different periods of time. By analyzing different time periods as outlined above, we gauge the stability of the factors across different periods of globalization. Moreover, our analysis follows the procedure of Ciccarelli and Mojon (2010) in assessing how much of the total variation in local inflation these global factors explain.

As a final step of the analysis, in Section 4 we gauge which economic variables are statistically associated with the estimated factors. As candidate variables, we include the VIX as a gauge of market volatility; the dollar index, which measures the value of the US dollar relative to a basket of foreign currencies; a commodity price index, as a gauge of the global role of tradable inputs; and the US federal funds rate, as a gauge of the global importance of major central banks' monetary policies. We obtain these data from the database of the St. Louis Fed (FRED). A natural variable to explain comovement in global inflation also lies in the presence of GVCs. We therefore also relate the share of local inflation in a country explained by one of the two factors from the factor analysis to three measures of GVCs developed in Bems and Johnson (2017). These are the foreign value added (FVA) embodied in a country's exports over the total value added and the domestic value added (DVX) of a country that is embodied in the exports of other countries, or, respectively, the sum of these two variables. We obtain these data from Casella et al. (2019). We complement these variables with a measure of global commodity prices, the primary commodity price index compiled by the IMF.

3 Global Inflation and Its Common Factors

The first and second factors of inflation show several pronounced characteristics over different periods of globalization: Independently of the period considered, the first factor appears relatively similar and explains a lot of variation. The second factor is unstable across periods and is associated with some common country-specific dynamics. Its global relevance, moreover, depends on the specific samples used.

We show these results by analyzing three periods: The first includes all available countries starting in 1960, the second those available after 1980, and the third focuses on the recent wave of globalization that accelerated after 2000. Table 1 describes the sample of countries included in each period, reflecting the unbalanced nature of the panel. The US and the western OECD countries are included in all samples. Figure 2 plots both first and second factors alongside average inflation in AEs and EMDEs for the countries in the sample from 2000 until June 2023. Figure 5 in the Appendix shows the results for the samples that start in 1980 and 1960.



Figure 2: Factors and Average Inflation for Different Samples

Notes: The panels in the figure show monthly inflation minus the sample average for the average advanced economies (AEs, left panel) and the average emerging market and developing economies (EMDEs, right panel) over different periods of time. Each panel includes the first and second factors estimated with the panel of countries for each period of time. Table 1 contains the underlying list of countries.

A first result that emerges is that the first factor appears relatively similar across all three periods for the common sample period, capturing common global inflation dynamics, even if the sample of countries increases over time. As is evident, the first factor was trending up before the global financial crisis and then decreased during the crisis, in line with its global reach. It is also evident that the first factor increased after the pandemic, in line with the experience that the increase in inflation that started in 2021 was common across countries.

A strong correlation between inflation and the first factor is characteristic for AEs. Inflation in EMDEs is more volatile. Notably, higher inflation before the global financial crisis is present in both groups of countries. This pre-GFC increase in inflation coincides with an increase in the first factor. However, for EMDEs this increase in inflation is relatively small compared to the inflation rates prevalent in the early 2000s. Similarly, after the GFC, advanced economies did not experience a big moderation in inflation relative to their average, which seems to have influenced the first global factor that remains stable until March 2020. The subsequent inflation surge in 2021 is very similar in magnitude, and simultaneous, for both AEs and EMDEs.

With respect to the second factor, we observe only a very weak correlation with inflation. The factor has a negative correlation with trend inflation in the early 2000s, especially for EMDEs, but remains flat for the rest, in contrast to the average variation that occurred in AEs and EMDEs between the GFC and the end of the sample in 2023.

In Figure 5 in the Appendix we also plot the first and second factors and average inflation for advanced economies for earlier periods in the data.⁶ The first factor shows marked increases in the 1970s in line with the common oil price shocks during that period. This experience was followed by a general decline in the 1980s. Global inflation dynamics captured by the first factor also tend to be more closely in line with the inflation dynamics in AEs in these earlier periods.

When it comes to the second factor, more nuanced results emerge. The second factor changes significantly depending on the sample under consideration. For example, when estimated based on the longest sample that starts in 1960, it declines during the recent post-pandemic surge of inflation. But when estimated on more recent samples, it increases or is relatively flat during this period, as shown in Figure 2. The likely reason for this result is that the second factor mostly explains the inflation experiences of countries with regional inflation trajectories and hence is more unstable than the first factor depending on periods and countries considered.

⁶We don't plot the average inflation of EMDEs for earlier periods, as those averages would be heavily influenced by hyperinflation experiences in the late 80s, and early 90s.

These two results on the first two factors are also reflected in the quantitative importance that they bear in explaining the variation in inflation. Table 1 shows how much of the inflation experience of each country each factor explains (as measured by R^2). The first factor dominantly explains the local inflation experience of most AEs. This holds true across different periods of globalization. For EMDEs, the importance of the first factor changes over time. For example, its importance is low for countries such as Peru or Argentina in the 1960s, but increases in later periods.

Countries 1st Comp 2nd Comp		1960-2023		1980	-2023	2000-2023		
Australia 0.078 0.004 0.062 0.017 0.077 0.033 Austria 0.617 0.120 0.583 0.002 0.868 0.003 Canada 0.332 0.033 0.881 0.023 0.698 0.000 Denmark 0.732 0.033 0.881 0.023 0.868 0.004 Finland 0.311 0.035 0.893 0.004 0.868 0.004 France 0.887 0.008 0.926 0.017 0.883 0.002 Germany 0.543 0.027 0.827 0.051 0.630 0.094 Italy 0.901 0.002 0.555 0.000 0.211 0.208 Luxembourg 0.740 0.445 0.688 0.077 0.812 0.002 New Zealand 0.691 0.225 0.623 0.733 0.788 0.006 Norway 0.775 0.032 0.776 0.032 0.778 0.032 Netzerland	Countries	1st Comp	2nd Comp	1st Comp	2nd Comp	1st Comp	2nd Comp	
Austria 0.617 0.120 0.583 0.002 0.881 0.036 Belgium 0.762 0.661 0.701 0.890 0.033 0.689 0.006 Canada 0.839 0.001 0.839 0.013 0.686 0.000 France 0.887 0.008 0.833 0.004 0.886 0.002 Grence 0.587 0.008 0.926 0.017 0.883 0.002 Grence 0.587 0.230 0.718 0.211 0.675 0.029 Ireland 0.841 0.027 0.827 0.051 0.630 0.094 Japan 0.474 0.142 0.602 0.000 0.201 0.208 Netherlands 0.416 0.223 0.768 0.002 0.399 0.010 Norway 0.702 0.000 0.740 0.002 0.399 0.010 Norway 0.779 0.030 0.774 0.038 0.778 0.036 Sp	Australia	0.768	0.004	0.662	0.015	0.737	0.034	
Belgium 0.772 0.061 0.701 0.080 0.881 0.026 Canada 0.332 0.033 0.881 0.023 0.098 0.000 Finland 0.311 0.035 0.893 0.004 0.868 0.004 France 0.887 0.008 0.926 0.017 0.883 0.002 Germany 0.543 0.027 0.827 0.000 0.911 0.630 0.094 Italy 0.901 0.002 0.555 0.000 0.912 0.000 Japan 0.474 0.445 0.688 0.077 0.812 0.002 New Zealand 0.691 0.025 0.625 0.023 0.739 0.010 Portugal 0.777 0.058 0.714 0.078 0.778 0.022 Sweden 0.771 0.004 0.822 0.009 0.804 0.021 United Kingdom 0.775 0.032 0.776 0.010 0.714 0.161	Austria	0.617	0.120	0.583	0.092	0.868	0.033	
Canada 0.839 0.013 0.698 0.000 Demmark 0.732 0.033 0.881 0.023 0.896 0.000 Finland 0.811 0.035 0.893 0.004 0.886 0.004 Greace 0.587 0.008 0.926 0.017 0.888 0.002 Germany 0.543 0.084 0.512 0.088 0.855 0.043 Grecce 0.587 0.230 0.718 0.211 0.675 0.029 Ireland 0.841 0.027 0.827 0.001 0.912 0.000 Japan 0.744 0.142 0.602 0.000 0.311 0.208 Luxembourg 0.702 0.000 0.740 0.002 0.399 0.010 Nerway 0.702 0.000 0.221 0.008 0.775 0.022 Sweizerland 0.483 0.022 0.658 0.031 0.673 0.020 United Kingdom 0.775 0.032	Belgium	0.762	0.061	0 701	0.080	0.851	0.026	
Demmark 0.732 0.033 0.881 0.023 0.896 0.000 France 0.887 0.008 0.926 0.017 0.883 0.002 Germany 0.543 0.084 0.512 0.088 0.855 0.043 Greece 0.587 0.230 0.718 0.211 0.675 0.029 Ireland 0.841 0.002 0.955 0.000 0.201 0.208 Japan 0.474 0.142 0.602 0.000 0.201 0.208 Netwerlands 0.691 0.025 0.625 0.023 0.739 0.006 Nerway 0.707 0.616 0.223 0.778 0.336 Spain 0.777 0.608 0.714 0.778 0.336 Spain 0.777 0.603 0.768 0.778 0.022 Sweden 0.771 0.004 822 0.009 0.804 0.027 Switzerland 0.475 0.322 0.766 <td< td=""><td>Canada</td><td>0.839</td><td>0.000</td><td>0.839</td><td>0.013</td><td>0.698</td><td>0.000</td></td<>	Canada	0.839	0.000	0.839	0.013	0.698	0.000	
Finland 0.811 0.035 0.893 0.004 0.888 0.004 France 0.887 0.008 0.926 0.017 0.888 0.003 Germany 0.543 0.084 0.512 0.088 0.855 0.003 Greee 0.587 0.230 0.718 0.211 0.675 0.004 Japan 0.474 0.142 0.602 0.000 0.312 0.002 Netherlands 0.691 0.025 0.662 0.000 0.310 0.208 Netwerlands 0.691 0.022 0.778 0.006 0.740 0.002 0.339 0.006 Norway 0.707 0.005 0.714 0.002 0.339 0.022 Sweden 0.771 0.004 0.822 0.009 0.804 0.027 Swetzerland 0.775 0.032 0.776 0.010 0.714 0.156 United Kingdom 0.775 0.032 0.776 0.010 0.714 0.15	Denmark	0.732	0.033	0.881	0.023	0.896	0.000	
France 0.887 0.008 0.926 0.017 0.883 0.002 Germany 0.543 0.084 0.512 0.083 0.855 0.043 Greece 0.587 0.230 0.718 0.211 0.675 0.029 Ireland 0.841 0.027 0.857 0.000 0.201 0.208 Lay 0.901 0.002 0.955 0.000 0.201 0.208 Laxembourg 0.474 0.416 0.223 0.768 0.000 Netwezaland 0.691 0.025 0.625 0.023 0.739 0.006 Nerway 0.777 0.608 0.714 0.078 0.778 0.036 Spain 0.777 0.608 0.714 0.778 0.022 Sweden 0.771 0.004 822 0.009 0.804 0.027 Switzerland 0.483 0.022 0.668 0.031 0.714 0.156 United Kingdom 0.775 0.032	Finland	0.811	0.035	0.893	0.004	0.868	0.004	
Germany 0.543 0.084 0.512 0.088 0.435 0.043 Greece 0.587 0.230 0.718 0.211 0.675 0.029 Ireland 0.841 0.027 0.827 0.001 0.912 0.000 Japan 0.474 0.142 0.602 0.000 0.211 0.208 Luxembourg 0.740 0.446 0.223 0.758 0.000 Netherlands 0.416 0.223 0.758 0.000 Nerway 0.702 0.000 0.741 0.075 0.778 0.035 Spain 0.779 0.000 0.921 0.008 0.775 0.022 Sweden 0.771 0.004 0.821 0.008 0.775 0.022 United Kingdom 0.775 0.032 0.766 0.031 0.673 0.020 United States 0.701 0.668 0.031 0.714 0.151 Cole 0.360 0.077 0.586 0.220 <td>France</td> <td>0.887</td> <td>0.008</td> <td>0.926</td> <td>0.017</td> <td>0.883</td> <td>0.002</td>	France	0.887	0.008	0.926	0.017	0.883	0.002	
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	Germany	0.543	0.084	0.512	0.088	0.855	0.043	
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Greece	0.587	0.230	0.718	0.211	0.675	0.029	
Italy 0.901 0.002 0.955 0.000 0.212 0.000 Luxembourg 0.740 0.045 0.688 0.077 0.812 0.002 Netherlands 0.416 0.223 0.768 0.000 New Zealand 0.691 0.025 0.625 0.023 0.778 0.006 New Zealand 0.757 0.058 0.714 0.078 0.778 0.036 Swaden 0.779 0.000 0.921 0.008 0.755 0.022 Sweizerland 0.483 0.022 0.658 0.031 0.673 0.020 United Kingdom 0.775 0.032 0.776 0.010 0.714 0.156 United States 0.702 0.001 0.715 0.032 0.707 0.003 Colombia 0.424 0.174 0.481 0.320 0.462 0.121 Hong Kong SAR 0.358 0.073 0.463 0.042 0.010 1.121 Hong Kong SAR <td< td=""><td>Ireland</td><td>0.841</td><td>0.027</td><td>0.827</td><td>0.051</td><td>0.630</td><td>0.094</td></td<>	Ireland	0.841	0.027	0.827	0.051	0.630	0.094	
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	Italy	0.901	0.002	0.955	0.000	0.912	0.000	
Luxembourg 0.740 0.045 0.688 0.077 0.812 0.002 New Zealand 0.691 0.025 0.625 0.023 0.768 0.000 Norway 0.702 0.000 0.740 0.002 0.399 0.010 Portugal 0.757 0.058 0.714 0.078 0.778 0.036 Sweden 0.779 0.000 0.921 0.009 0.804 0.027 Sweten 0.771 0.004 0.822 0.009 0.804 0.027 Sweten 0.771 0.004 0.822 0.009 0.804 0.027 Switzerland 0.483 0.022 0.658 0.031 0.714 0.156 United States 0.702 0.001 0.715 0.032 0.707 0.003 Colombia 0.424 0.174 0.431 0.320 0.462 0.121 Hong Kong SAR 0.358 0.073 0.463 0.066 0.020 0.033	Japan	0.474	0.142	0.602	0.000	0.201	0.208	
Netherlands 0.416 0.223 0.768 0.000 New Zealand 0.691 0.025 0.625 0.023 0.739 0.006 Norway 0.702 0.000 0.740 0.002 0.339 0.016 Portugal 0.757 0.000 0.921 0.008 0.755 0.022 Sweden 0.771 0.004 0.822 0.009 0.804 0.020 United Kingdom 0.775 0.032 0.776 0.010 0.714 0.156 United States 0.702 0.001 0.715 0.035 0.707 0.066 Colombia 0.424 0.174 0.431 0.320 0.462 0.121 Hong Kong SAR 0.358 0.073 0.463 0.866 0.000 0.616 Ireiand 0.135 0.002 0.227 0.013 0.054 0.033 Ireiand 0.358 0.021 0.626 0.370 0.017 Sorad 0.017 0.554	Luxembourg	0.740	0.045	0.688	0.077	0.812	0.002	
New Zealand 0.691 0.025 0.625 0.023 0.739 0.006 Portugal 0.757 0.058 0.714 0.078 0.036 Spain 0.779 0.000 0.921 0.008 0.755 0.022 Sweden 0.771 0.004 0.822 0.009 0.804 0.027 Switzerland 0.483 0.022 0.658 0.031 0.673 0.020 United Kingdom 0.775 0.032 0.776 0.010 0.714 0.156 United States 0.702 0.001 0.715 0.032 0.767 0.032 Colombia 0.424 0.174 0.431 0.320 0.462 0.121 Hong Kong SAR 0.358 0.077 0.586 0.220 0.629 0.060 Ireal 0.210 0.134 0.334 0.042 0.319 0.001 India 0.135 0.002 0.227 0.013 0.55 0.017 Suta 0	Netherlands			0.416	0.223	0.768	0.000	
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	New Zealand	0.691	0.025	0.625	0.023	0.739	0.006	
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Norway	0.702	0.000	0.740	0.002	0.399	0.010	
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Portugal	0.757	0.058	0.714	0.078	0.778	0.036	
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	Spain	0.779	0.000	0.921	0.008	0.755	0.022	
Switzerland 0.483 0.022 0.658 0.031 0.673 0.020 United Kingdom 0.775 0.032 0.776 0.010 0.714 0.156 United States 0.702 0.001 0.715 0.035 0.707 0.003 Argentina 0.010 0.052 0.019 0.661 0.204 0.161 Chile 0.360 0.077 0.586 0.220 0.660 0.600 Colombia 0.424 0.174 0.431 0.320 0.662 0.121 Horg Kong SAR 0.358 0.073 0.463 0.004 0.230 0.001 India 0.135 0.0024 0.676 0.0042 0.334 0.032 0.001 Peru 0.006 0.092 0.244 0.132 0.001 South Africa 0.353 0.379 0.444 0.332 0.100 0.006 Indonesia 0.0554 $0.$	Sweden	0.771	0.004	0.822	0.009	0.804	0.027	
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Switzerland	0.483	0.022	0.658	0.031	0.673	0.020	
	United Kingdom	0.775	0.032	0.776	0.010	0.714	0.156	
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$	United States	0.702	0.001	0.715	0.035	0.707	0.003	
$\begin{array}{c cccc} {\rm Chile} & 0.360 & 0.077 & 0.586 & 0.220 & 0.629 & 0.060 \\ {\rm Colombia} & 0.424 & 0.174 & 0.431 & 0.320 & 0.462 & 0.121 \\ {\rm Hong Kong SAR} & 0.358 & 0.073 & 0.463 & 0.086 & 0.000 & 0.616 \\ {\rm Iceland} & 0.670 & 0.024 & 0.676 & 0.004 & 0.230 & 0.001 \\ {\rm India} & 0.135 & 0.002 & 0.227 & 0.013 & 0.054 & 0.033 \\ {\rm Israel} & 0.210 & 0.134 & 0.334 & 0.042 & 0.319 & 0.000 \\ {\rm Peru} & 0.006 & 0.092 & 0.024 & 0.127 & 0.452 & 0.091 \\ {\rm Prilippines} & 0.255 & 0.044 & 0.241 & 0.062 & 0.370 & 0.017 \\ {\rm South Africa} & 0.353 & 0.379 & 0.444 & 0.332 & 0.100 & 0.000 \\ {\rm Indonesia} & & 0.007 & 0.006 & 0.016 & 0.070 \\ {\rm Korea} & & 0.354 & 0.027 & 0.170 & 0.060 \\ {\rm Singapore} & & 0.388 & 0.126 & 0.525 & 0.096 \\ {\rm Thailand} & & 0.354 & 0.027 & 0.170 & 0.060 \\ {\rm Singapore} & & 0.388 & 0.126 & 0.525 & 0.096 \\ {\rm Thailand} & & 0.469 & 0.005 & 0.406 & 0.013 \\ {\rm Turkey} & & 0.147 & 0.095 & 0.458 & 0.247 \\ {\rm Algeria} & & & & 0.127 & 0.167 \\ {\rm Brazil} & & & & & 0.127 & 0.167 \\ {\rm Brazil} & & & & & & 0.014 & 0.149 \\ {\rm Croatia} & & & & & & & 0.014 & 0.149 \\ {\rm Croatia} & & & & & & & & & 0.020 & 0.071 \\ {\rm Bulgaria} & & & & & & & & & & & & & & & & & & &$	Argentina	0.010	0.052	0.019	0.061	0.204	0.161	
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Chile	0.360	0.077	0.586	0.220	0.629	0.060	
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	Colombia	0.424	0.174	0.431	0.320	0.462	0.121	
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Hong Kong SAR	0.358	0.073	0.463	0.086	0.000	0.616	
$\begin{array}{c ccccc} \mbox{India} & 0.135 & 0.002 & 0.227 & 0.013 & 0.054 & 0.033 \\ \mbox{Israel} & 0.210 & 0.134 & 0.334 & 0.042 & 0.319 & 0.000 \\ \mbox{Peru} & 0.006 & 0.092 & 0.024 & 0.127 & 0.452 & 0.091 \\ \mbox{Philippines} & 0.255 & 0.044 & 0.241 & 0.062 & 0.370 & 0.017 \\ \mbox{South Africa} & 0.353 & 0.379 & 0.444 & 0.332 & 0.100 & 0.000 \\ \mbox{Indonesia} & & 0.007 & 0.006 & 0.016 & 0.070 \\ \mbox{Korea} & & 0.554 & 0.002 & 0.550 & 0.018 \\ \mbox{Malaysia} & & 0.354 & 0.027 & 0.170 & 0.060 \\ \mbox{Singapore} & & 0.388 & 0.126 & 0.525 & 0.096 \\ \mbox{Thailand} & & 0.469 & 0.005 & 0.406 & 0.013 \\ \mbox{Turkey} & & 0.147 & 0.095 & 0.458 & 0.247 \\ \mbox{Algeria} & & & 0.129 & 0.167 \\ \mbox{Brazil} & & & 0.129 & 0.167 \\ \mbox{Brazil} & & & 0.129 & 0.167 \\ \mbox{Brazil} & & & 0.079 & 0.026 \\ \mbox{Coratia} & & & 0.014 & 0.149 \\ \mbox{Croatia} & & & 0.799 & 0.026 \\ \mbox{Czechia} & & & 0.799 & 0.026 \\ \mbox{Czechia} & & & 0.799 & 0.026 \\ \mbox{Czechia} & & & 0.718 & 0.016 \\ \mbox{Hungary} & & & 0.708 & 0.035 \\ \mbox{Latvia} & & & 0.751 & 0.134 \\ \mbox{North Macedonia} & & & 0.751 & 0.134 \\ \mbox{North Macedonia} & & & 0.014 & 0.214 \\ \mbox{Serbia} & & & 0.000 \\ \mbox{Poland} & & & 0.014 & 0.214 \\ \mbox{Serbia} & & & 0.000 \\ \mbox{Poland} & & & 0.000 \\ \mbox{Poland} & & & 0.004 & 0.735 \\ \mbox{South Republic} & & & 0.601 & 0.172 \\ \mbox{Sovenia} & & & 0.601 & 0.172 \\ \mbox{Average Sample 1980} & 0.579 & 0.064 & 0.613 & 0.074 & 0.595 & 0.058 \\ \mbox{Average Sample 1980} & 0.567 & 0.069 & 0.557 & 0.062 \\ \mbox{Average Sample 1980} & 0.579 & 0.064 & 0.567 & 0.069 & 0.557 & 0.062 \\ \mbox{Average Sample 1980} & 0.579 & 0.064 & 0.567 & 0.069 & 0.557 & 0.062 \\ \mbox{Average Sample 1980} & 0.579 & 0.064 & 0.567 & 0.069 & 0.557 & 0.062 \\ \mbox{Average Sample 1980} & 0.567 & 0.069 & 0.557 & 0.062 \\ \mbox{Average Sample 1980} & 0.567 & 0.069 & 0.557 & 0.062 \\ \mbox{Average Sample 1980} & 0.567 & 0.069 & 0.557 & 0.062 \\ \mbox{Average Sample 1980} & 0.567 & 0.069 & 0.557 & 0.062 \\ \mbox{Average Sample 1980} & 0.567 & 0.069 & 0.557 & $	Iceland	0.670	0.024	0.676	0.004	0.230	0.001	
	India	0.135	0.002	0.227	0.013	0.054	0.033	
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Israel	0.210	0.134	0.334	0.042	0.319	0.000	
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Peru	0.006	0.092	0.024	0.127	0.452	0.091	
South Africa 0.353 0.379 0.444 0.332 0.100 0.000 Indonesia 0.007 0.006 0.016 0.070 Korea 0.554 0.002 0.550 0.018 Malaysia 0.354 0.027 0.170 0.060 Singapore 0.388 0.126 0.525 0.096 Thailand 0.469 0.005 0.406 0.013 Turkey 0.147 0.095 0.458 0.247 Algeria 0.129 0.167 0.390 0.167 Brazil 0.020 0.071 0.020 0.071 Bulgaria 0.147 0.095 0.458 0.247 Croatia 0.014 0.149 0.020 0.071 Bulgaria 0.014 0.149 0.020 0.071 Cyprus 0.708 0.038 0.039 0.035 Czechia 0.851 0.016 0.918 0.016 Hungary 0.708 0.032	Philippines	0.255	0.044	0.241	0.062	0.370	0.017	
Indonesia 0.007 0.006 0.016 0.070 Korea 0.554 0.002 0.550 0.018 Malaysia 0.354 0.027 0.170 0.066 Singapore 0.388 0.126 0.525 0.096 Thailand 0.469 0.005 0.406 0.013 Turkey 0.147 0.095 0.458 0.247 Algeria 0.129 0.167 Brazil 0.020 0.071 Bulgaria 0.014 0.149 Croatia 0.014 0.149 Cyprus 0.709 0.026 Czechia 0.709 0.026 Estonia 0.708 0.035 Latvia 0.751 0.134 North Macedonia 0.749 0.014 Romania 0.014 0.219 Serbia 0.014 0.219 Slovenia 0.755 0.034 Average Sample 1960 0.722 0.044 0.735 0.049	South Africa	0.353	0.379	0.444	0.332	0.100	0.000	
Korea 0.554 0.002 0.550 0.018 Malaysia 0.354 0.027 0.170 0.060 Singapore 0.388 0.126 0.525 0.096 Thailand 0.469 0.005 0.406 0.013 Turkey 0.147 0.095 0.458 0.247 Algeria 0.147 0.095 0.458 0.247 Bulgaria 0.147 0.095 0.458 0.247 Croatia 0.736 0.039 0.071 0.149 Croatia 0.736 0.039 0.014 0.149 Croatia 0.709 0.266 0.709 0.026 Czechia 0.709 0.026 0.709 0.026 Czechia 0.709 0.035 0.035 0.458 Latvia 0.708 0.335 0.014 0.145 North Macedonia 0.751 0.134 0.014 Romania 0.014 0.145 0.755 Saudi Arabia 0.014 0.145 0.755 Slovenia 0.060 0.527 0.064 0.613 Average Sample 1960 0.722 0.044 0.735 0.049 0.737 Average Sample 1980 0.567 0.069 0.557 0.062	Indonesia			0.007	0.006	0.016	0.070	
Malaysia 0.354 0.027 0.170 0.060 Singapore 0.388 0.126 0.525 0.096 Thailand 0.469 0.005 0.406 0.013 Turkey 0.147 0.095 0.458 0.247 Algeria 0.129 0.167 0.020 0.071 Bulgaria 0.736 0.029 0.014 Croatia 0.014 0.149 Croatia 0.014 0.149 Cyprus 0.709 0.026 Czechia 0.851 0.016 Estonia 0.788 0.039 Lithuania 0.751 0.134 North Macedonia 0.749 0.014 Poland 0.749 0.014 Serbia 0.068 0.527 Slovak Republic 0.601 0.172 Slovak Republic 0.664 0.633 Average Sample 1960 0.579 0.064 0.613 0.074 0.555 Average Sample 1980 0.567 0.062 0.567 0.062	Korea			0.554	0.002	0.550	0.018	
Singapore 0.388 0.126 0.525 0.096 Thailand 0.469 0.005 0.406 0.013 Turkey 0.147 0.095 0.458 0.247 Algeria 0.129 0.167 Brazil 0.020 0.071 Bulgaria 0.014 0.129 0.167 Croatia 0.014 0.149 0.039 Croatia 0.014 0.149 0.026 Czechia 0.882 0.001 0.266 Estonia 0.918 0.016 0.458 Hungary 0.709 0.026 0.266 Latvia 0.708 0.035 0.035 Latvia 0.6684 0.039 0.145 0.755 Saudi Arabia 0.749 0.014 0.214 Serbia 0.014 0.214 0.014 0.214 Serbia 0.0601 0.172 0.064 0.613 0.074 0.595 0.054 Average Sample 1960 0.579 0.064 0.613 0.074 0.595 0.058 Ave	Malaysia			0.354	0.027	0.170	0.060	
Thailand 0.469 0.005 0.406 0.013 Turkey 0.147 0.095 0.458 0.247 Algeria 0.129 0.167 Brazil 0.020 0.071 Bulgaria 0.736 0.039 China 0.014 0.149 Croatia 0.882 0.001 Cyprus 0.709 0.026 Czechia 0.881 0.016 Estonia 0.918 0.016 Hungary 0.708 0.035 Latvia 0.708 0.039 Lithuania 0.751 0.134 North Macedonia 0.832 0.000 Poland 0.014 0.214 Serbia 0.014 0.214 Slovenia 0.6601 0.172 Slovenia 0.600 0.319 Average Sample 1960 0.579 0.064 0.613 0.074 0.555 0.058 Average Sample 1980 0.567 0.062 0.567 0.062	Singapore			0.388	0.126	0.525	0.096	
Turkey 0.147 0.095 0.458 0.247 Algeria 0.129 0.167 Brazil 0.020 0.071 Bulgaria 0.736 0.039 China 0.014 0.149 Croatia 0.882 0.001 Cyprus 0.709 0.026 Czechia 0.851 0.016 Estonia 0.708 0.039 Latvia 0.708 0.035 Latvia 0.751 0.134 North Macedonia 0.751 0.134 North Macedonia 0.014 0.214 Serbia 0.014 0.214 Slovak Republic 0.668 0.527 Slovak Republic 0.600 0.319 Average OECD 1960 0.722 0.044 0.735 0.049 Average Sample 1960 0.579 0.064 0.613 0.074 0.595 Average Sample 1980 0.567 0.069 0.557 0.058	Thailand			0.469	0.005	0.406	0.013	
Algeria 0.129 0.167 Brazil 0.020 0.071 Bulgaria 0.14 0.149 Croatia 0.882 0.001 Cyprus 0.736 0.039 Cyprus 0.709 0.026 Czechia 0.851 0.016 Estonia 0.918 0.016 Hungary 0.708 0.035 Latvia 0.6684 0.039 Lithuania 0.751 0.134 North Macedonia 0.749 0.014 Romania 0.014 0.214 Serbia 0.060 0.527 Slovenia 0.600 0.319 Average Sample 1960 0.579 0.064 0.613 0.074 0.595 0.058 Average Sample 1980 0.567 0.069 0.557 0.062	Turkey			0.147	0.095	0.458	0.247	
Brazil 0.020 0.071 Bulgaria 0.736 0.039 China 0.014 0.149 Croatia 0.882 0.001 Cyprus 0.709 0.026 Czechia 0.81 0.016 Estonia 0.918 0.016 Hungary 0.708 0.035 Latvia 0.751 0.134 North Macedonia 0.751 0.134 North Macedonia 0.749 0.014 Serbia 0.014 0.2145 Serbia 0.014 0.214 Slovenia 0.600 0.122 Average Sample 1960 0.579 0.064 0.613 0.074 0.595 Average Sample 1980 0.567 0.069 0.557 0.062	Algeria					0.129	0.167	
Bulgaria 0.736 0.039 China 0.014 0.149 Croatia 0.882 0.001 Cyprus 0.709 0.026 Czechia 0.851 0.016 Estonia 0.918 0.016 Hungary 0.708 0.039 Latvia 0.684 0.039 Lithuania 0.751 0.134 North Macedonia 0.832 0.000 Poland 0.749 0.014 Romania 0.014 0.214 Serbia 0.014 0.214 Slovenia 0.600 0.319 Average OECD 1960 0.722 0.044 0.735 0.049 0.737 0.034 Average Sample 1960 0.579 0.064 0.613 0.074 0.595 0.058 Average Sample 1980 0.567 0.069 0.557 0.062	Brazil					0.020	0.071	
Chna 0.014 0.149 Croatia 0.882 0.001 Cyprus 0.709 0.026 Czechia 0.851 0.016 Estonia 0.918 0.016 Hungary 0.684 0.035 Latvia 0.684 0.039 Lithuania 0.751 0.134 North Macedonia 0.832 0.000 Poland 0.749 0.014 Romania 0.145 0.755 Saudi Arabia 0.014 0.214 Serbia 0.060 0.527 Slovenia 0.600 0.319 Average OECD 1960 0.722 0.044 0.735 0.049 0.737 Average Sample 1960 0.579 0.064 0.613 0.074 0.595 0.058 Average Sample 1980 0.567 0.069 0.557 0.062	Bulgaria					0.736	0.039	
Croatia 0.882 0.001 Cyprus 0.709 0.026 Czechia 0.851 0.016 Estonia 0.918 0.016 Hungary 0.708 0.035 Latvia 0.684 0.039 Lithuania 0.751 0.134 North Macedonia 0.779 0.014 Romania 0.145 0.755 Saudi Arabia 0.014 0.214 Serbia 0.601 0.172 Slovenia 0.600 0.319 Average Sample 1960 0.579 0.064 0.613 0.074 0.595 0.058 Average Sample 1980 0.567 0.069 0.557 0.062	China					0.014	0.149	
Cyprus 0.709 0.026 Czechia 0.851 0.016 Estonia 0.918 0.016 Hungary 0.708 0.035 Latvia 0.684 0.039 Lithuania 0.751 0.134 North Macedonia 0.832 0.000 Poland 0.749 0.014 Romania 0.145 0.755 Saudi Arabia 0.014 0.214 Serbia 0.068 0.527 Slovenia 0.600 0.319 Average OECD 1960 0.722 0.044 0.735 0.049 Average Sample 1960 0.579 0.064 0.613 0.074 0.595 Average Sample 1980 0.567 0.069 0.557 0.058	Croatia					0.882	0.001	
Czecnia 0.851 0.016 Estonia 0.918 0.016 Hungary 0.708 0.035 Latvia 0.684 0.039 Lithuania 0.751 0.134 North Macedonia 0.749 0.014 Romania 0.145 0.755 Saudi Arabia 0.014 0.214 Serbia 0.068 0.527 Slovenia 0.600 0.319 Average OECD 1960 0.722 0.044 0.735 0.049 0.737 0.034 Average Sample 1960 0.579 0.064 0.613 0.074 0.595 0.058 Average Sample 1980 0.567 0.069 0.557 0.062 0.546 0.000	Cyprus					0.709	0.026	
Lstonia 0.918 0.016 Hungary 0.708 0.035 Latvia 0.684 0.039 Lithuania 0.751 0.134 North Macedonia 0.832 0.000 Poland 0.749 0.014 Romania 0.145 0.755 Saudi Arabia 0.014 0.214 Serbia 0.6601 0.172 Slovenia 0.600 0.319 Average OECD 1960 0.722 0.044 0.735 Average Sample 1960 0.579 0.064 0.613 Average Sample 1980 0.567 0.062 Average Sample 2000 0.567 0.062	Czechia					0.851	0.016	
Hungary 0.708 0.035 Latvia 0.684 0.039 Lithuania 0.751 0.134 North Macedonia 0.832 0.000 Poland 0.749 0.014 Romania 0.145 0.755 Saudi Arabia 0.014 0.214 Serbia 0.068 0.527 Slovenia 0.601 0.172 Average OECD 1960 0.722 0.044 0.735 0.049 0.737 0.034 Average Sample 1960 0.579 0.064 0.613 0.074 0.595 0.058 Average Sample 1980 0.567 0.069 0.557 0.062 0.568	Estonia					0.918	0.016	
Latvia 0.084 0.039 Lithuania 0.751 0.134 North Macedonia 0.832 0.000 Poland 0.749 0.014 Romania 0.145 0.755 Saudi Arabia 0.014 0.214 Serbia 0.608 0.527 Slovenia 0.600 0.319 Average OECD 1960 0.722 0.044 0.735 0.049 0.737 0.034 Average Sample 1960 0.579 0.064 0.613 0.074 0.595 0.058 Average Sample 1980 0.567 0.069 0.557 0.062	Hungary					0.708	0.035	
Littmana 0.751 0.134 North Macedonia 0.832 0.000 Poland 0.749 0.014 Romania 0.145 0.755 Saudi Arabia 0.014 0.214 Serbia 0.068 0.527 Slovenia 0.600 0.319 Average OECD 1960 0.722 0.044 0.735 0.049 0.737 0.034 Average Sample 1960 0.579 0.064 0.613 0.074 0.595 0.058 Average Sample 2000 0.567 0.069 0.557 0.062 0.567 0.069	Latvia					0.084	0.039	
North Macedonia 0.832 0.000 Poland 0.749 0.014 Romania 0.145 0.755 Saudi Arabia 0.014 0.214 Serbia 0.068 0.527 Slovenia 0.601 0.172 Average OECD 1960 0.722 0.044 0.735 0.049 0.737 0.034 Average Sample 1960 0.579 0.064 0.613 0.074 0.595 0.058 Average Sample 1980 0.567 0.069 0.557 0.062 0.900	Lithuania					0.751	0.134	
Romania 0.145 0.014 Romania 0.145 0.755 Saudi Arabia 0.014 0.214 Serbia 0.068 0.527 Slovenia 0.600 0.319 Average OECD 1960 0.722 0.044 0.735 0.049 0.737 0.034 Average Sample 1960 0.579 0.064 0.613 0.074 0.595 0.058 Average Sample 1980 0.5667 0.069 0.557 0.062 0.566 0.000	Dol					0.632	0.000	
Average Sample 1980 0.745 0.755 Average Sample 1980 0.014 0.214 0.214 Outer Sample 1980 0.6613 0.068 0.527 Outer Sample 1980 0.604 0.613 0.074 0.334 Average Sample 1980 0.567 0.069 0.557 0.062	Poland					0.749	0.014	
Saturi Arabia 0.014 0.114 Serbia 0.068 0.527 Slovak Republic 0.601 0.172 Slovenia 0.600 0.319 Average OECD 1960 0.579 0.064 0.613 0.074 0.595 0.058 Average Sample 1980 0.567 0.069 0.557 0.062 Average Sample 2000 0 0.567 0.062 0.567 0.062	Romania Saudi Anabia					0.145	0.700	
Serona 0.008 0.227 Slovak Republic 0.601 0.172 Slovenia 0.600 0.319 Average OECD 1960 0.722 0.044 0.735 0.049 0.737 0.034 Average Sample 1960 0.579 0.064 0.613 0.074 0.595 0.058 Average Sample 2000 0.567 0.069 0.557 0.062	Saudi Arabia					0.014	0.214	
Slovenia 0.001 0.172 Average OECD 1960 0.722 0.044 0.735 0.049 0.737 0.034 Average Sample 1960 0.579 0.064 0.613 0.074 0.595 0.058 Average Sample 1980 0.5667 0.069 0.557 0.062 Average Sample 2000 0 0.546 0.000 0.546 0.000	Slovak Popublic					0.000	0.027	
Average GECD 1960 0.722 0.044 0.735 0.049 0.737 0.034 Average Sample 1960 0.579 0.064 0.613 0.074 0.595 0.058 Average Sample 1980 0.567 0.069 0.557 0.062 Average Sample 2000 0.567 0.069 0.557 0.062	Slovenia					0.600	0.172	
Average Sample 1960 0.722 0.044 0.735 0.049 0.737 0.034 Average Sample 1960 0.579 0.064 0.613 0.074 0.595 0.058 Average Sample 1980 0.567 0.069 0.557 0.062 Average Sample 2000 0.546 0.000 0.546 0.000	Augrage OECD 1060	0.722	0.044	0.725	0.040	0.000	0.024	
Average Sample 1980 0.013 0.014 0.393 0.058 Average Sample 2000 0.567 0.069 0.557 0.062	Average Sample 1060	0.722	0.044	0.730	0.049	0.737	0.034	
Average Sample 2000 0.567 0.009 0.567 0.002	Average Sample 1900	0.015	0.004	0.567	0.069	0.557	0.055	
	Average Sample 2000			0.007	0.003	0.546	0.090	

Table 1: Share of Local Inflation Explained by Global Factors

Note: This table shows the proportion of variance that is explained by the factor for each country. As in Ciccarelli and Mojon (2010), the estimates capture the R^2 of a regression of the national inflation rate on a factor and a constant.

By contrast, the second factor generally does not explain a large share of the variation in inflation. Notably, however, it tends to account for more of the local inflation experiences that are not well-explained by the first factor, such as for Argentina or Hong Kong. For example, Hong Kong experienced large negative inflation rates at the beginning of the 2000s and relatively stable positive inflation thereafter. The second factor captures this experience well, especially in the most recent, post-2000 sample. But the second factor also describes well countries that experienced significant disinflation in the early 2000s, such as Romania. At the same time, the first factor explains very little of the variation of such inflation experiences. The next section explores variables that can account for these results in an economic rather than a statistical sense.

We next investigate the share of local *core* inflation that is explained by global factors. While most central banks set their objective target looking at CPI headline inflation (or other measures of aggregate inflation), others also consider core inflation, a measure of inflation that removes volatile factors such as food and/or energy. Our above result that the first factor significantly explains local inflation aligns with such a focus on core inflation. In a world where global factors explain most of the fluctuation in inflation, it might be informative to remove some of the most tradable components that cannot necessarily be controlled by domestic monetary policy.

Interestingly, global factors also explain a big proportion of countries' core inflation. To establish this fact, we repeat our analysis, but for a subset of countries where we have access to a measure of core inflation. Table 2 shows the list of countries we use for this specific analysis. Figure 6 in the Appendix plots the core factor along with AEs, EMDEs and US core inflation. Our previous findings continue to apply for core inflation, which Table 2 shows: The first global core inflation factor explains a big proportion of core inflation. Table 8 in the Appendix shows that the results in terms of correlation with global variables also continue to apply to this core global factor. Finally, Figure 3 shows a high correlation between a country's headline inflation explained by the first global factor and the country's core inflation explained by the first global factor.

Our results thus indicate that a global factor explains a big proportion of inflation across countries, even when we consider less tradable goods (as captured by the core inflation measure). The next section discusses how trade integration can help rationalize these findings.

Table 2:	Share	of	Local	Core	Inflation	Exp	lained	by	Global	Factors
----------	-------	----	-------	------	-----------	-----	--------	----	--------	---------

Countries	First Factor	Second Factor
Austria	0.750	0.112
Belgium	0.778	0.058
Canada	0.548	0.049
Chile	0.489	0.123
Colombia	0.449	0.115
Czechia	0.475	0.250
Denmark	0.793	0.011
Estonia	0.791	0.007
Finland	0.660	0.019
France	0.623	0.011
Germany	0.623	0.152
Greece	0.413	0.230
Hungary	0.712	0.036
Iceland	0.250	0.007
Ireland	0.342	0.162
Israel	0.229	0.003
Italy	0.677	0.156
Japan	0.057	0.227
Korea, Rep.	0.444	0.098
Latvia	0.292	0.001
Lithuania	0.415	0.309
Luxembourg	0.646	0.006
Netherlands	0.645	0.005
Norway	0.316	0.129
Poland	0.588	0.000
Portugal	0.709	0.133
Slovak Republic	0.567	0.145
Slovenia	0.486	0.291
Spain	0.589	0.213
Sweden	0.714	0.060
Switzerland	0.463	0.020
Turkiye	0.481	0.073
United Kingdom	0.483	0.286
United States	0.623	0.090
Average	0.533	0.105

Note: This table shows the proportion of variance that is explained by the factor for each country. As in Ciccarelli and Mojon (2010), the estimates capture the R^2 of a regression of the national core inflation rate on a factor and a constant.

Figure 3: Relationship All and Core Factor



Notes: The panels in the figure show the correlation between the share of local CPI inflation (x axis) or core inflation (y axis) explained by the global factor. Each dot represents a country. Those shares are calculated for the sample of countries in Table 2 from 2000 to 2023. The slope for the first factor is 0.778 (robust standard errors of 0.075, R^2 of 0.702) and for the second factor it is 0.684 (robust standard errors of 0.190, R^2 of 0.325).

4 Economic Forces Behind Inflation Factors

Economically, what is behind the relevance of these statistical factors? This section presents evidence for the importance of several global economic variables for the two factors and also highlights the importance of GVCs for each factor.

4.1 The correlates of global inflation

First, our analysis shows that several global economic variables are associated with the first and second factors generated by the factor model. These variables include the dollar index, the US federal funds rate, the primary commodity price index and the VIX index. They capture various shocks that drive costs and lead to inflationary pressure, but they also capture the role of US monetary policy and general market volatility.

		First	Factor			Second	Factor	
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
VIX	0.004				0.011***			
	(0.003)				(0.004)			
Dollar Index		0.036^{***}				-0.017^{***}		
		(0.002)				(0.003)		
Commodity Price Index		. ,	0.008^{***}			. ,	0.008^{***}	
			(0.001)				(0.001)	
Federal Funds Rate				0.217^{***}				-0.048^{***}
				(0.006)				(0.014)
Constant	-0.476^{***}	-3.332***	-1.615^{***}	-1.015***	-0.034	1.461^{***}	-0.478^{*}	0.133^{***}
	(0.057)	(0.197)	(0.123)	(0.025)	(0.085)	(0.223)	(0.097)	(0.051)
Observations	402	480	246	497	402	480	246	497
R-squared	0.003	0.271	0.356	0.771	0.010	0.073	0.421	0.047

Table 3: Factors and Other Global Variables

Note: This table shows the results of a regression where the factor is the dependent variable. The independent variables are described in the first column of the table. For Columns (1) to (4) the dependent variable is the first factor. For Columns (5) to (8) it is the second factor. VIX is the market expectation of near-term volatility according to the Chicago Board Options Exchange, Inc. Dollar Index is the trade-weighted US dollar index (goods only) that goes from 1973 to 2019. Commodity Price Index is the primary commodity price index built by the IMF that represents the benchmark commodity prices on the global market. The Federal Fund Rate is the Federal Reserve policy rates from 1960 to 2023. Robust standard errors in parenthesis.

Regarding the first factor, we find a positive association with all four variables as Table 3 shows. The first factor has a positive, statistically significant correlation with the global dollar index. This association could indicate that changes in the US dollar translate into costs in other countries, but it could also capture inflationary pressures in the US. The positive correlation with the commodity price index reflects common movements in input cost. The first factor also strongly and statistically significantly correlates with the US federal funds rate. This association encapsulates global inflationary pressures and a policy reaction from the Federal Reserve. While the first factor is also positively correlated with the global VIX, its relationship does not have statistical significance.

Regarding the second factor, we find a statistically significant relationship with all four variables. But unlike in the case of the first factor, the VIX now takes on statistical significance. Since the second factor is usually correlated more with inflation dynamics for countries that are not affected by global inflation trends (as discussed in the previous section), this finding suggests that the inflationary pressures of these countries are in particular related to higher global uncertainty.

4.2 US, China and global inflation

We next explore how the factors correlate with inflation in the US and China. In order to do so, we take the whole time series and run a regression where the dependent variable is either the first or second factor and the independent variable is US inflation, or, respectively, Chinese inflation. First, in terms of US inflation, Figure 4 shows inflation in the US and the factors for the different sample periods. A high correlation between US inflation and the first global factor is apparent.



Figure 4: Factors and US Inflation for Different Samples

Notes: Figures show monthly inflation for the US minus the sample average over different periods of time and the first and second factors estimated with the panel of countries for each period of time. Table 7 contains the underlying list of countries.

Interestingly, the recent increase in the first factor after the COVID pandemic appears delayed compared to the rise of inflation in the US. Notably, its recent decline also appears delayed compared to the evolution of US inflation. Additionally, US inflation declined faster than the first factor at the end of the Great Inflation likely due to the monetary policy tightening of Paul Volcker.

A set of regression specifications allows us to quantify the relationship further, in particular by allowing us to also include as dynamic control onemonth lagged US inflation. Table 4 shows estimation results for the sample from 1960 to 2023. Tables 10 and 11 in the Appendix show estimates for other sample periods, with generally similar findings.

		First Factor		Second Factor			
Inflation US	0.297^{***}		0.024	-0.009		0.019	
	(0.008)		(0.052)	(0.012)		(0.084)	
Lagged Inflation US		0.300^{***}	0.276^{***}		-0.009	-0.028	
		(0.007)	(0.051)		(0.013)	(0.085)	
Constant	-1.123^{***}	-1.130***	-1.131***	0.035	0.036	0.036	
	(0.028)	(0.028)	(0.028)	(0.051)	(0.051)	(0.051)	
Sample	1960-2023	1960-2023	1960-2023	1960-2023	1960-2023	1960-2023	
Observations	761	761	761	761	761	761	
R-squared	0.717	0.729	0.729	0.001	0.001	0.001	

Table 4: First and Second Factors and US inflation

Note: This table shows the results of a regression where the factor is the dependent variable. The independent variables are described in the first column of the table. For Columns (1) to (3) the dependent variable is the first factor. For Columns (4) to (6) it is the second factor. Inflation US is annual US inflation with a monthly frequency. Lagged Inflation US is the one-month lagged annual US inflation with a monthly frequency. The sample goes from February 1960 to June 2023. Robust standard errors in parenthesis.

Results indicate that there is a strong positive relationship between the first factor and US inflation. Also, US inflation appears to account for a big proportion in the variation of the first factor, as seen by the R^2 summarized in Table 1. Importantly, while lagged inflation also explains a big proportion of the first factor, when both current and lagged inflation are included, lagged inflation becomes statistically significant. This finding confirms the suggestive interpretation in the above discussion of Figure 4, as US inflation seems to lead the first factor. In terms of the second factor, the estimated coefficient is small and not statistically significant. US inflation does not explain much of the variation in this factor.

Interestingly, this is not true for other relevant global countries. Table 5 shows the relationship between Chinese inflation and the first and second factors; Figure 7 in the Appendix plots the factors alongside Chinese inflation. While the estimated coefficients are positive and statistically different, the explanatory power of Chinese inflation is relatively small. Additionally, if

anything, as the figure illustrates, the second factor seems to be more in line with Chinese inflation than the first factor, mainly due to the increase in inflation in the early 2000s. At the same time, the second factor seems to be important for some other countries as well, as Table 1 shows. For example, it explains more than 75% of the inflation variation in Romania. But Romania experienced a disinflation in the early 2000s, not inflation like China. Therefore, how to interpret this second factor is a bit unclear. Thus, while Chinese inflation might have some relationship to the second factor, its relevance is relatively small, especially compared to the high correlation between the first global factor and US inflation.

		First Factor		Second Factor				
Inflation China	0.061^{***}		-0.044	0.203^{***}		-0.003		
	(0.021)		(0.080)	(0.028)		(0.096)		
Lagged Inflation China		0.070^{***}	0.112		0.208^{***}	0.212^{**}		
		(0.021)	(0.081)		(0.027)	(0.095)		
Constant	-0.131	-0.153*	-0.148*	-0.437^{***}	-0.439^{***}	-0.438^{***}		
	(0.081)	(0.079)	(0.080)	(0.102)	(0.100)	(0.101)		
Sample	2000-2023	2000-2023	2000-2023	2000-2023	2000-2023	2000-2023		
Observations	282	281	281	282	281	281		
R-squared	0.014	0.019	0.019	0.160	0.174	0.174		

Table 5: First and Second Factors and Inflation in China

Note: This table shows the results of a regression where the factor is the dependent variable. The independent variables are described in the first column of the table. For Columns (1) to (3) the dependent variable is the first factor. For Columns (4) to (6) it is the second factor. Inflation China is the annual inflation in China with a monthly frequency. Lagged Inflation China is the one-month lagged annual inflation in China with a monthly frequency. The sample goes from January 2000 to June 2023. Robust standard errors in parenthesis.

4.3 Relationship to GVC linkages

A particular variable of interest in understanding global inflation dynamics is structural, pertaining to the impact of the global value chains. How much global factors affect local inflation should be a function of how integrated local economies are with international trade networks and by the extent to which global shocks can affect local costs and price setting (see, for example, Gilchrist and Zakrajsek (2019); Auer, Levchenko, and Sauré (2019); Amiti et al. (2023)). What is the impact of global value chains on the importance of the *factors* underlying inflation dynamics?

In order to see how relevant international integration is in this context,

we estimate the strength of the relationship between the share of inflation explained by our two factors and the share of value added in each country that is related to global value chains. Specifically, we estimate the following specification:

$$s_i^{factor} = \alpha + \beta_1 GVC_i + \varepsilon_i, \tag{2}$$

where s_i^{factor} denotes the share of local inflation in country *i* explained by a factor. The variable GVC_i is defined by the average global value added participation index of country *i* between 2000 and 2018 over the total value added. This index is equal to $GVC_i = FVA_i + DVX_i$, where FVA_i denotes the foreign value added (FVA) embodied in country *i*'s exports over the total value added, or equivalently, the value added of a country's exports that is explained by imported intermediate inputs. In turn, DVX_i denotes the domestic value added (DVX) of country *i* that is embodied in the exports of other countries, relative to the total value added, or how much country *i* adds to the exports of a third country. Our analysis also regresses s_i^{factor} on each of these factors separately so as to gauge its individual contributions.

						~ ~			
		First	Factor		Second Factor				
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	
FVA	0.817^{***}		0.897^{***}		0.091		0.077		
	(0.274)		(0.278)		(0.136)		(0.126)		
DVX		-0.546	0.265			-0.115	-0.046		
		(0.416)	(0.464)			(0.233)	(0.209)		
GVC				0.869^{***}				0.071	
				(0.274)				(0.121)	
Const	0.278***	0.723***	0.169	-0.008	0.052	0.118	0.070	0.0036	
	(0.100)	(0.133)	(0.201)	(0.188)	(0.039)	(0.079)	(0.075)	(0.068)	
Obs	55	55	55	55	55	55	55	55	
R2	0.175	0.026	0.180	0.145	0.010	0.005	0.010	0.004	

Table 6: Global Value Added and Global Inflation

Note: This table shows the results of a regression where the share of local inflation explained by global factors is the dependent variable, as in Table 1. The independent variables are described in the first column of the table. For Columns (1) to (4) the dependent variable is the first factor. For Columns (5) to (8) it is the second factor. FVA denotes a country's exports over the total value added, or equivalently, the value added of a country's exports that is explained by imported intermediate inputs over the total value added. DVX denotes domestic value added over the total value added that is embodied in the exports of other countries. GVC contains the sum of FVA and DVX. Robust standard errors in parenthesis.

Two main results emerge, as summarized in Table 6. First, countries with a higher share of their local inflation explained by the first global factor also have a higher share of their foreign value added (i.e., their local production uses more imported intermediate inputs). This result tends to originate from the information encapsulated by the FVA variable, which measures how much of a country's exports are explained by value added from imports, in other words, its export dependence on imported intermediate inputs. This result therefore suggests that the capacity of the first factor to explain local inflation is related to the extent of the trade integration of countries, especially trade integration of their supply chains. Notably, the DVX variable by itself does not contribute greatly to the overall explanatory power of GVCs for the first factor while the FVA variable carries most of the information. Thus, the correlation of our GVC variable with the first factor mainly stems from the extent to which foreign value added in general is embodied in a country's exports over the total value added.

The second result that emerges in the context of GVCs relates to the sec-

ond factor. Simply, this second factor does not bear any strong relationship to GVCs. In general, there is no strong, statistically significant correlation with this factor. In addition, the R^2 tends to be low which suggests that little of this factor is explained by global trade factors.

Table 9 in the Appendix shows that this finding also applies to core inflation. Overall we see that the results for aggregate inflation apply in a very similar way to the findings using core inflation.

5 Conclusion

This chapter has examined the international co-movement of inflation using a dynamic factor model between 1960 and 2023, extending the seminal work of Ciccarelli and Mojon (2010) to include the time since the global financial crisis, including the post-pandemic inflation surge. The first factor of inflation explains a high proportion of inflation across countries and periods of globalization, especially in advanced economies. The second factor captures experiences in a smaller number of countries, mostly developing economies. During the post-pandemic inflation surge, US inflation was leading both factors while in earlier periods global inflation dynamics were more synchronized with the US inflation experience. Inflation in China is mildly associated with the second factor.

Economically, what is behind the relevance of these statistical factors? Global economic variables such as the dollar index, the US federal funds rate, and commodity prices are associated with the first and second factors generated by the factor model while the VIX is more relevant for the second factor. Moreover, GVCs play an important role: GVC measures account for a significant proportion of the share of local inflation dynamics that are explained by the first factor of our factor analysis.

The results in this chapter indicate that global factors are relevant for explaining local inflation. This result is particularly important for countries that depend more heavily on imported inputs as the importance of GVCs indicates. Importantly, this finding also holds true for inflation components that are not imported, i.e. core inflation. Taken together, these findings highlight the challenges that local monetary authorities face in a globally integrated economy, such as the choice of appropriate policy tools to stabilize prices when global prices are quite volatile.

References

- Amiti, Mary, Sebastian Heise, Fatih Karahan, and Ayşegül Şahin. 2023. "Inflation Strikes Back: The Role of Import Competition and the Labor Market." Tech. rep., National Bureau of Economic Research. URL https://doi.org/10.3386/w31211.
- Ascari, Guido and Luca Fosso. 2024. "The international dimension of trend inflation." *Journal of International Economics* 148:103896. URL https: //doi.org/10.1016/j.jinteco.2024.103896.
- Auer, Raphael, Claudio Borio, and Andrew Filardo. 2017. "The globalisation of inflation: the growing importance of global value chains." *BIS Working Papers* (602). URL https://www.bis.org/publ/work602.htm.
- Auer, Raphael and Andreas M. Fischer. 2010. "The effect of low-wage import competition on US inflationary pressure." *Journal of Monetary Economics* 57 (4):491–503. URL https://doi.org/10.1016/j.jmoneco.2010.02.007.
- Auer, Raphael and Aaron Mehrotra. 2014. "Trade linkages and the globalization of inflation in Asia and the Pacific." Journal of International Money and Finance 49:Part A. URL https://doi.org/10.1016/j.jimonfin.2014.05. 008.
- Auer, Raphael A, Andrei A Levchenko, and Philip Sauré. 2019. "International inflation spillovers through input linkages." *Review of Economics* and Statistics 101 (3):507–521. URL https://doi.org/10.1162/rest_a_00781.
- Bandera, Nicolò, Lauren Barnes, Matthieu Chavaz, Silvana Tenreyro, and Lukas von dem Berge. 2023. "Monetary policy in the face of supply shocks: The role of inflation expectations." In ECB Forum on Central Banking. 26–28.
- Bems, Rudolfs and Robert C Johnson. 2017. "Demand for value added and value-added exchange rates." American Economic Journal: Macroeconomics 9 (4):45–90. URL https://doi.org/10.1257/mac.20150216.
- Bergholt, Drago, Francesco Furlanetto, and Etienne Vaccaro-Grange. 2023. "Did monetary policy kill the Phillips curve? Some simple arithmetics." Norges Bank Working Papers (2). URL https://hdl.handle.net/11250/ 3050868.

- Bianchi, F. and A. Civelli. 2015. "Globalization and inflation: evidence from a time varying VAR." *Review of Economic Dynamics* 18 (2):406–433. URL https://doi.org/10.1016/j.red.2014.07.004.
- Blinder, Alan S. 1981. "Monetary Accommodation of Supply Shocks under Rational Expectations." Journal of Money, Credit and Banking 13 (4):425–438. URL https://doi.org/10.2307/1991933.
- Borio, Claudio and Andrew Filardo. 2007. "Globalisation and inflation: New cross-country evidence on the global determinants of domestic inflation." *BIS Working Papers* (227). URL https://www.bis.org/publ/work227. htm.
- Bruno, Valentina and Hyun Song Shin. 2015. "Capital flows and the risktaking channel of monetary policy." Journal of Monetary Economics 71:119–132. URL https://doi.org/10.1016/j.jmoneco.2014.11.011.
- Casella, Bruno, Richard Bolwijn, Daniel Moran, and Keiichiro Kanemoto. 2019. "Improving the analysis of global value chains: the UNCTAD-Eora Database." *Transnational Corporations* 26 (3):115–142. URL https://doi. org/10.18356/3aad0f6a-en.
- Ciccarelli, Matteo and Benoit Mojon. 2010. "Global inflation." *The Review* of *Economics and Statistics* 92 (3):524–535. URL https://doi.org/10.1162/ REST_a_00008.
- De Gregorio, José. 2012. "Commodity prices, monetary policy, and inflation." *IMF Economic Review* 60 (4):600–633. URL https://doi.org/10. 1057/imfer.2012.15.
- Degasperi, Riccardo, Simon Hong, and Giovanni Ricco. 2020. "The Global Transmission of U.S. Monetary Policy." URL https://cepr.org/ publications/dp14533.
- Egorov, Konstantin and Dmitry Mukhin. 2023. "Optimal Policy under Dollar Pricing." American Economic Review 113 (7):1783–1824. URL https: //doi.org/10.1257/aer.20200636.
- Forbes, Kristin. 2019. "Inflation dynamics: Dead, dormant, or determined abroad?" Brookings Papers on Economic Activity :257–338URL https://www.jstor.org/stable/26996612.

- Forni, Mario, Marc Hallin, Marco Lippi, and Lucrezia Reichlin. 2000. "The generalized dynamic-factor model: Identification and estimation." *Review* of Economics and statistics 82 (4):540–554. URL https://doi.org/10.1162/ 003465300559037.
- Frache, Serafin, Rodrigo Lluberas, Mathieu Pedemonte, and Javier Turen. 2023. "The Transmission of International Monetary Policy Shocks on Firms' Expectations." URL https://www.doi.org/10.26509/ frbc-wp-202301.
- Gilchrist, Simon, Raphael Schoenle, Jae Sim, and Egon Zakrajšek. 2017. "Inflation dynamics during the financial crisis." *American Economic Review* 107 (3):785–823. URL https://doi.org/10.1257/aer.20150248.
- Gilchrist, Simon, Raphael Schoenle, Jae Sim, and Egon Zakrajšek. 2023. "Financial heterogeneity and monetary union." Journal of Monetary Economics 139:21–40. URL https://doi.org/10.1016/j.jmoneco.2023.06.002.
- Gilchrist, Simon and Egon Zakrajsek. 2019. "Trade Exposure and the Evolution of Inflation Dynamics." Working Papers Central Bank of Chile 849, Central Bank of Chile. URL https://ideas.repec.org/p/chb/bcchwp/849. html.
- Gopinath, Gita, Emine Boz, Camila Casas, Federico J Díez, Pierre-Olivier Gourinchas, and Mikkel Plagborg-Møller. 2020. "Dominant currency paradigm." American Economic Review 110 (3):677–719. URL https: //doi.org/10.1257/aer.20171201.
- Ha, Jongrim, M Ayhan Kose, and Franziska Ohnsorge. 2019. "Global inflation synchronization." CEPR Discussion Paper 13600. URL https: //doi.org/10.2139/ssrn.3352919.
- Ha, Jongrim, M. Ayhan Kose, Franziska Ohnsorge, and Hakan Yilmazkuday. 2023. "What Explains Global Inflation." Policy Research Working Paper 10648, World Bank, Washington, DC. URL https://doi.org/10.2139/ssrn. 4668343.
- Kalemli-Özcan, Şebnem. 2019. "US monetary policy and international risk spillovers." Tech. rep., National Bureau of Economic Research. URL https: //doi.org/10.3386/w26297.

- Kalemli-Ozcan, Şebnem and Filiz D Unsal. 2024. "Global transmission of FED Hikes: The role of policy credibility and balance sheets." Tech. rep., National Bureau of Economic Research. URL https://doi.org/10.3386/ w32329.
- McLeay, Michael and Silvana Tenreyro. 2020. "Optimal Inflation and the Identification of the Phillips Curve." NBER Macroeconomics Annual 34:199–255. URL https://doi.org/10.1086/707181.
- Miranda-Agrippino, Silvia and Hélene Rey. 2020. "US monetary policy and the global financial cycle." *The Review of Economic Studies* 87 (6):2754– 2776. URL https://doi.org/10.1093/restud/rdaa019.
- Mukhin, Dmitry. 2022. "An Equilibrium Model of the International Price System." American Economic Review 112 (2):650–88. URL https://doi. org/10.1257/aer.20181550.
- Mumtaz, Haroon and Paolo Surico. 2009. "The transmission of international shocks: a factor-augmented VAR approach." *Journal of Money, Credit and Banking* 41 (s1):71–100. URL https://doi.org/10.1111/j.1538-4616. 2008.00199.x.

———. 2012. "Evolving international inflation dynamics: world and country-specific factors." *Journal of the European Economic Association* 10 (4):716–734. URL https://doi.org/10.1111/j.1542-4774.2012.01068.x.

- Obstfeld, Maurice, Jonathan D Ostry, and Mahvash S Qureshi. 2019. "A tie that binds: Revisiting the trilemma in emerging market economies." *Review of Economics and Statistics* 101 (2):279–293. URL https://doi.org/10.1162/rest_a_00740.
- Peersman, Gert. 2022. "International food commodity prices and missing (dis) inflation in the euro area." *Review of Economics and Statistics* 104 (1):85–100. URL https://doi.org/10.1162/rest_a_00939.
- Rey, Hélène. 2015. Dilemma not trilemma: The global financial cycle and monetary policy independence. Global Dimensions of Unconventional Monetary Policy. Federal Reserve Bank of Kansas City.

- Stock, James H and Mark W Watson. 2002. "Macroeconomic forecasting using diffusion indexes." Journal of Business & Economic Statistics 20 (2):147–162. URL https://doi.org/10.1198/073500102317351921.
- Wynne, Mark. 2012. "Five years of research on globalization and monetary policy: what have we learned?" Federal Reserve Bank of Dallas Annual Report URL https://fraser.stlouisfed.org/title/6346/item/606583? start_page=4.

Appendix

_

EMDEs	AEs				
Algeria	Australia	New Zealand			
Argentina	Austria	Norway			
Brazil	Belgium	Poland			
Bulgaria	Canada	Portugal			
Chile	Croatia	Romania			
China	Cyprus	Slovak Republic			
Colombia	Czechia	Slovenia			
Hong Kong	Denmark	Spain			
India	Estonia	Sweden			
Indonesia	Finland	Switzerland			
Israel	France	United Kingdom			
Malaysia	Germany	United States			
North Macedonia	Greece				
Peru	Hungary				
Philippines	Iceland				
Saudi Arabia	Ireland				
Serbia	Italy				
Singapore	Japan				
South Africa	Latvia				
South Korea	Lithuania				
Thailand	Luxembourg				
Türkiye	Netherlands				

Table 7: List of Countries

Note: This table lists the countries used for the main results of this paper. It also shows the country classification used in Figure 4. EMDEs are "emerging market and developing economies" and AEs are "advanced economies." We classify countries based on the latest country classification in the BIS Annual Economic Reports (see https://www.bis.org/statistics/country_groupings.pdf).

		First	Factor			Second	l Factor	
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Vix	0.031^{***}				0.017^{***}			
	(0.006)				(0.006)			
Dollar Index		0.018^{***}				0.040^{***}		
		(0.003)				(0.004)		
Commodity Price Index		. ,	0.011^{***}			. ,	-0.011***	
			(0.002)				(0.001)	
Federal Funds Rate				0.128^{***}			. ,	0.274^{***}
				(0.010)				(0.018)
Constant	-0.617^{***}	-1.774^{***}	-1.562^{***}	-0.445***	-0.337**	-1.197^{***}	1.219^{***}	-0.331***
	(0.121)	(0.235)	(0.245)	(0.031)	(0.130)	(0.184)	(0.207)	(0.056)
Observations	282	240	246	257	282	240	246	257
R-squared	0.064	0.179	0.171	0.253	0.020	0.226	0.281	0.319

Table 8: Core Factors and Other Global Variables

Note: This table shows the results of a regression where the factor is the dependent variable. The independent variables are described in the first column of the table. For Columns (1) to (4) the dependent variable is the first factor for the core inflation sample. For Columns (5) to (8) it is the second factor for the core inflation sample. VIX is the market expectation of near term volatility according to the Chicago Board Options Exchange, Inc. Dollar Index is the real broad dollar index (goods only) that goes from 1973 to 2019. Commodity Price Index is the primary commodity price index built by the IMF; that is, the value represents the benchmark prices that are representative of the global market. The Federal Fund Rate is the Federal Reserve policy rates from 1960 to 2023. Robust standard errors in parenthesis.





Notes: Figures show monthly inflation for advanced economies on average over different periods of time and the first and second factors estimated with the panel of countries for each period of time. Table 1 contains the underlying list of countries.

Figure 6: Core Factor and Core Inflation



Notes: Figures show monthly core inflation minus the sample average for advanced economies (top left), emerging markets and developing economies (top right) and the US (bottom) since 2000 and the first and second factors estimated with the panel of countries for which we have core inflation. Table 2 contains the underlying list of countries.

		First F	actor		Second Factor					
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)		
FVA	0.548**		0.561**		-0.068		-0.080			
	(0.228)		(0.228)		(0.107)		(0.110)			
DVX		-0.373	0.053			0.010	-0.051			
		(0.318)	(0.342)			(0.188)	(0.205)			
GVC				0.514^{**}				-0.077		
				(0.227)				(0.106)		
Const	0.326***	0.647^{***}	0.305^{*}	0.182	0.131***	0.102*	0.151*	0.158**		
	(0.094)	(0.105)	(0.156)	(0.163)	(0.040)	(0.059)	(0.082)	(0.070)		
Obs	34	34	34	34	34	34	34	34		
R2	0.170	0.026	0.171	0.125	0.009	0.000	0.010	0.010		

Table 9: Global Value Added and Global Core Inflation

Note: This table shows the results of a regression where the share of local core inflation explained by global factors is the dependent variable, as in Table 2. The independent variables are described in the first column of the table. For Columns (1) to (4) the dependent variable is the first factor. For Columns (5) to (8) it is the second factor. FVA denotes a country's exports over the total value added, or equivalently, the value added of a country's exports that is explained by imported intermediate inputs over the total value added. DVX denotes domestic value added over the total value added that is embodied in the exports of other countries. GVC contains the sum of FVA and DVX. Robust standard errors in parenthesis.

Table 10: First and Second Factors and US Inflation

		First Factor		Second Factor		
Inflation US	0.336^{***}		-0.004	0.076^{***}		-0.015
	(0.009)		(0.060)	(0.014)		(0.099)
Lagged Inflation US		0.336^{***}	0.339^{***}		0.077^{***}	0.091
		(0.009)	(0.059)		(0.013)	(0.099)
Constant	-1.125^{***}	-1.130***	-1.129^{***}	-0.255^{***}	-0.258^{***}	-0.257^{***}
	(0.032)	(0.030)	(0.031)	(0.052)	(0.052)	(0.052)
Sample	1980-2023	1980-2023	1980-2023	1980-2023	1980-2023	1980-2023
Observations	522	522	522	522	522	522
R-squared	0.724	0.743	0.743	0.040	0.041	0.041

Note: This table shows the results of a regression where the factor is the dependent variable. The independent variables are described in the first column of the table. For Columns (1) to (3) the dependent variable is the first factor. For Columns (4) to (6) it is the second factor. Inflation US is annual US inflation with a monthly frequency. Lagged Inflation US is the one-month lagged annual US inflation with a monthly frequency. The sample goes from January 1980 to June 2023. Robust standard errors in parenthesis.

		First Factor		Second Factor			
Inflation US	0.465^{***}		0.008	0.030		-0.037	
	(0.030)		(0.067)	(0.024)		(0.102)	
Lagged Inflation US		0.480^{***}	0.473^{***}		0.033	0.069	
		(0.027)	(0.074)		(0.024)	(0.102)	
Constant	-1.184^{***}	-1.223^{***}	-1.224^{***}	-0.075	-0.084	-0.081	
	(0.066)	(0.062)	(0.062)	(0.073)	(0.072)	(0.073)	
Sample	2000-2023	2000-2023	2000-2023	2000-2023	2000-2023	2000-2023	
Observations	282	282	282	282	282	282	
R-squared	0.717	0.729	0.729	0.001	0.001	0.001	

Table 11: First and Second Factors and US inflation

Note: This table shows the results of a regression where the factor is the dependent variable. The independent variables are described in the first column of the table. For Columns (1) to (3) the dependent variable is the first factor. For Columns (4) to (6) it is the second factor. Inflation US is annual US inflation with a monthly frequency. Lagged Inflation US is the one-month lagged annual US inflation with a monthly frequency. The sample goes from January 2000 to June 2023. Robust standard errors in parenthesis.





Notes: The figure shows monthly headline inflation minus the sample average for China since 2000 and the first and second factors estimated with the panel of countries. Table 2 contains the underlying list of countries.

Previous volumes in this series

1188 May 2024	Finding a needle in a haystack: a machine learning framework for anomaly detection in payment systems	Ajit Desai, Anneke Kosse and Jacob Sharples
1187 May 2024	Nothing to hide? Gender and age differences in willingness to share data	Olivier Armantier, Sebastian Doerr, Jon Frost, Andreas Fuster and Kelly Shue
1186 May 2024	Unconditional convergence in the Mexican manufacturing sector (1988-2018)	Alex Rivadeneira
1185 May 2024	Allocative efficiency and the productivity slowdown	Lin Shao and Rongsheng Tang
1184 May 2024	Determinants of currency choice in cross- border bank loans	Lorenz Emter, Peter McQuade, Swapan-Kumar Pradhan and Martin Schmitz
1183 May 2024	Why DeFi lending? Evidence from Aave V2	Giulio Cornelli, Leonardo Gambacorta, Rodney Garratt and Alessio Reghezza
1182 April 2024	Reserve requirements as a financial stability instrument	Carlos Cantú, Rocio Gondo and Berenice Martinez
1181 April 2024	Synthetic controls with machine learning: application on the effect of labour deregulation on worker productivity in Brazil	Douglas K G Araujo
1180 April 2024	Incentive-Compatible Unemployment Reinsurance for the Euro Area	Alexander Karaivanov, Benoit Mojon, Luiz Pereira da Silva, Albert Pierres Tejada and Robert M Townsend
1179 April 2024	The impact of artificial intelligence on output and inflation	Iñaki Aldasoro, Sebastian Doerr, Leonardo Gambacorta and Daniel Rees
1178 April 2024	Finternet: the financial system for the future	Agustín Carstens and Nandan Nilekani
1177 March 2024	Pre-publication revisions of bank financial statements: a novel way to monitor banks?	Andre Guettler, Mahvish Naeem, Lars Norden and Bernardus F Nazar Van Doornik
1176 March 2024	The effect of Covid pension withdrawals and the Universal Guaranteed Pension on the income of future retirees in Chile	Carlos Madeira

All volumes are available on our website www.bis.org.